Author's response to reviews

Title: Identifying strategies to maximise recruitment and retention of practices and patients in a multicentre randomised controlled trial of an intervention to optimise secondary prevention for coronary heart disease in primary care.

Authors:

Claire S Leathem (claire.l@talktalk.net)
Margaret E Cupples (m.cuples@qub.ac.uk)
Mary C Byrne (mary.byrne@nuigalway.ie)
Mary O'Malley (srah2000@yahoo.com)
Ailish Houlihan (ailish@croi.ie)
Andrew W Murphy (andrew.murphy@nuigalway.ie)
Susan M Smith (susmith@tcd.ie)

Version: 5 Date: 15 June 2009

Author's response to reviews:

We have made the requested formatting changes (major and minor) in the attached version.